Dear Dr Grandjean,

Thank you for giving us the opportunity to revise our MS 1527554270912099.

We are submitting a new version of our manuscript and would like to inform that we have revised the tables and figures according to the instructions given to the authors and we have made the spelling and grammatical corrections you suggested.

We believe we have made all the necessary changes in the style and formatting of the manuscript as required for the editorial handling.

Yours sincerely,

Agnes Soares da Silva

Corresponding author